Author’s response to reviews

Title: Neuromuscular exercise reduces low back pain intensity and improves physical functioning in nursing duties among female healthcare workers; secondary analysis of a randomised controlled trial

Authors:

Anni Taulaniemi (annika.taulaniemi@ukkinstituutti.fi)
Markku Kankaanpää (markku.kankaanpaa@pshp.fi)
Kari Tokola (kari.tokola@ukkinstituutti.fi)
Jari Parkkari (jari.parkkari@ukkinstituutti.fi)
Jaana Suni (jaana.suni@ukkinstituutti.fi)

Version: 3 Date: 11 Jun 2019

Author’s response to reviews:

1. We note that you have indicated that your study is a secondary analysis of your previous RCT, in the Methods section of your manuscript. Please be sure to indicate your study design in both the title and in the Abstract of your manuscript.

Term "secondary analysis" has been added to the Title and to the Abstract.

2. As this is a secondary analysis, please remove the trial registration number from the abstract and only include it in the Methods section.

Trial registration number is included only in the Method section of the current manuscript.

3. Please include a Conclusions heading in the text of your manuscript, as both a Conclusion and a Discussion section should be provided. You may consider placing it before the last paragraph of the Discussion section.

Conclusion heading has been added.
4. Thank you for including the individual contributions of each author in the Authors' Contributions section. Please be sure to also specify how each author contributed to the secondary analysis of your study.

Each author's contribution to the current manuscript has been specified.

5. In the funding section, please also state the role of the funding body in the design of the study, collection, analysis, interpretation of data, and in writing the manuscript.

The role of the funding body has been completed.

6. Please proofread and ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes, colored text, or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

The revised manuscript is in its final form.